Author's response to reviews

Title: Screening and association testing of common coding variation in steroid hormone receptor co-activator and co-repressor genes in relation to breast cancer risk: the Multiethnic Cohort

Authors:

Christopher A Haiman (haiman@usc.edu)
Rachel R Garcia (garcia.rachel@gmail.com)
Chris Hsu (chrishsu@usc.edu)
Lucy Xia (lyxia@usc.edu)
Helen Ha (helenha@usc.edu)
Xin Sheng (xsheng@usc.edu)
Loic Le Marchand (loic@crch.hawaii.edu)
Laurence N Kolonel (larry@crch.hawaii.edu)
Brian E Henderson (behender@usc.edu)
Michael R Stallcup (stallcup@usc.edu)
Geoffrey L Greene (greene@uchicago.edu)
Michael F Press (press@usc.edu)

Version: 3 Date: 27 January 2009

Author's response to reviews: see over
January 27, 2009

Dr. Melissa Norton, MD
Editor-in-Chief
BMC Cancer

Dear Dr. Norton,

Please find attached our revised manuscript, “Screening and Association Testing of Common Coding Variation in Steroid Hormone Receptor Co-activator and Co-repressor Genes in Relation to Breast Cancer Risk: The Multiethnic Cohort”. We have re-labeled the “supplemental tables” to “additional files” as well as made the minor revisions requested.

Please direct all correspondence to Christopher Haiman.

Sincerely,

Christopher A. Haiman, Sc.D.
Associate Professor, Department of Preventive Medicine
Keck School of Medicine
University of Southern California
USC/Norris Comprehensive Cancer Center
Harylna Norris Research Tower
1450 Biggy Street, room 1504A
Los Angeles, California 90033
Phone: 323 442-7755; Fax: 323 442-7749
e-mail: haiman@usc.edu